Author’s response to reviews

Title: Evaluation of the content quality of websites for recurrent aphthous ulcers and oral lichen planus

Authors:

Xiaosheng Hu (xiaohudie0613@163.com)
Hui Pan (panhui213@163.com)
Wen Xiu He (wxhe1995@126.com)
Hong Hua (honghua1968@aliyun.com)
Zhimin Yan (yzhimin96@163.com)

Version: 2 Date: 13 Dec 2017

Author’s response to reviews:

Review 1

Thank you for the amended manuscript following my recommendation. There are still a number of small typos but overall this paper reads a lot better. I recommend some language corrections before being published. The followings are some minor errors P3 line 3 papular (not popular), plaque-like, atrophic, erosive line 7 rate of 1.1% (not 0.9% - please check the reference carefully)

1. Thank you for your revision. The mentioned language flaws have been corrected.

2. According to the reference, the rate of 1.1% was analyzed from all the included studies, while the rate of 0.9% was concluded from studies that used the World Health Organization-2003 diagnostic criteria. We originally chose the latter considering that it was more convincible with clear diagnostic criteria.